Author's response to reviews

Title: Factors associated with latent tuberculosis among asylum seekers in Switzerland: A cross-sectional study in Vaud County

Authors:

Apostolos Sarivalasis (apostolos.sarivalasis@yahoo.com)
Jean - Pierre Zellweger (zellwegerjp@swissonline.ch)
Mohamed Faouzi (mohamed.faouzi@chuv.ch)
Oscar Daher (oscar.daher@cssc.ch)
Charlotte Deslarzes (charlotte.deslarzes@hospvd.ch)
Patrick Bodenmann (patrick.bodenmann@hospvd.ch)

Version: 3 Date: 25 September 2012

Author's response to reviews: see over
Lausanne, the 29th of August 2012

About: Cover Letter of the revised manuscript

Dear Editor,

We thank you for having revised our manuscript. We have studied the associated editor comment and integrated his proposition. Please find in attachment:

1. Our answer to your comment.

2. The manuscript in a revised form, with all changes highlighted in color.

We hope that the text in the current form will be accepted for publication and await your decision.

With our best regards

On behalf of the authors

Dr. A. Sarivalasis
ANSWER TO THE COMMENT OF THE ASSOCIATED EDITOR

Editorial Board Comments

Figures 6 and 7 are essentially showing the same information, and the legend for figure 6 is not accurate (the ROC curve demonstrates different sensitivities/specificities at a range of different cutoffs, not just at a score of 13 and above). I would delete figure 7 and just label the point on Figure 6 corresponding to the sensitivity/specificity of a score of 13.

Response to the Editorial Board Comments

We thank the associated editor for his comment. We agree with his suggestion and deleted the figure 7 from the final manuscript. The figure 6 was labeled according to the suggestion of the associated editor and the manuscript was updated to match those changes.